Author's response to reviews

Title: Barriers and facilitators to primary care for people with mental health and/or substance use issues: A qualitative study

Authors:

Lori E Ross (l.ross@utoronto.ca)
Simone Vigod (simone.vigod@wchospital.ca)
Jessica Wishart (jess.wishart@gmail.com)
Myera Waese (myera.zoey@gmail.com)
Jason (Dean) Spence (jasondeanspence@yahoo.ca)
Jason Oliver (jasonaaronoliver1981@gmail.com)
Jennifer Chambers (jennifer.wayward@gmail.com)
Scott Anderson (sanderson@catie.ca)
Roslyn Shields (roslyn.shields@camh.ca)

Version: 2 Date: 21 May 2015

Author's response to reviews:

Please note that in this version, we have added line numbers and page numbers, corrected the format of the author contributions section, and indicated that all authors have read and approved the final manuscript.